Author's response to reviews

Title: Clinical presentation of human African trypanosomiasis in Zambia links with varying virulent strains of human trypanosomes: Two case reports.

Authors:

Victor Mwanakasale (vicsale@zamtel.zm)
Peter Songolo (songolop@zm.afro.who.int)
Olusegun Babaniyi (babaniyio@zm.afro.who.int)
Pere Simarro (SimarroP@who.int)

Version: 2 Date: 4 July 2013

Author's response to reviews: see over
4th July 2013

The Editor
Journal of Medical Case Reports

Dear Sir/Madam

Re: Submission of amended version of the two case reports for possible publication.

On behalf of my co-authors I wish to submit an amended version of our two case reports entitled “Clinical presentation of human African trypanosomiasis in Zambia links with varying virulent strains of human trypanosomes: Two case reports” for possible publication in the Journal of Medical Case reports. We have since added a statement on the Consent section explaining the reasons why we couldn’t obtain consent from the parents of patient 2 that died for Case 2.

We look forward to a favourable response from your office.

Yours Faithfully

COPPERBELT UNIVERSITY

Dr Victor Mwanakasale
CORRESPONDING AUTHOR